Author's response to reviews

Title: Lower energy expenditures in infants from obese biological mothers

Authors:

Russell Rising (Russell_rising@yahoo.com)
Fima Lifshitz (DrLfshitz@aol.com)

Version: 4 Date: 15 May 2008

Author's response to reviews:

The changes you are requesting were minor.

1. Affiliations were rewritten in full and included city and country as indicated. We did include a state since it would be difficult for readers to know where the city is in the United States.

2. I listed the corresponding author as myself and included an email address

3. Changed "Introduction" to "Background" as requested